Answering reviewers

Reviewer’s code: 06058880

SPECIFIC COMMENTS TO AUTHORS
Prognostic nutrition index initially proposed to predict the immunogenicity and risk of gastrointestinal surgery through albumin and lymphocytes, but it was found in the later stage that it was associated with the prognosis of many tumors. At present, it is considered to have predictive significance in the prognosis of lung cancer and it is believed that low prognostic nutrition index is related to poor prognosis. In this study, the authors investigated the predictive value of preoperative nutritional index for biological recurrence in high/extremely high-risk prostate cancer patients after radical surgery. The study is very well designed. The data in results are very interesting, and well discussed. Minor comments: 1. There are some minor language polishing which should be corrected. 2. The references list should be edited according to the journal’s guideline. 3. The tables and figures should be moved to the end of the text.

Reply1: The language in the manuscript was revised again.
Reply2: References have been revised in the required format.
Reply3: Forms and drawings have been moved to the end of the manuscript.
Reviewer’s code: 06058848

SPECIFIC COMMENTS TO AUTHORS
This is an interesting study of the value of preoperative nutritional index in the prognosis evaluation of high-risk/extremely high-risk prostate cancer patients after radical prostatectomy. The manuscript is very well written. A minor editing is required before acceptance.

Reply1: The manuscript has been checked and edited again.